Author's response to reviews

Title: Rationale and Design of the CAPAMIS Study: Effectiveness of Pneumococcal Vaccination against Community-Acquired Pneumonia, Acute Myocardial Infarction and Stroke

Authors:

Angel Vila-Corcoles (avila.tarte.ics@gencat.cat)
Immaculada Hospital-Guardiola (ihospitalg.tarte.ics@gencat.cat)
Olga Ochoa-Gondar (ochoa.tarte.ics@gencat.cat)
Cinta de Diego (mcdiego.tarte.ics@gencat.cat)
Elisabet Salsench (eli.ss@terra.es)
Xavier Raga (xraga@xarxatecla.cat)
Cruz M Fuentes-Bellido (cfuentes.tarte.ics@gencat.cat)

Version: 3 Date: 15 January 2010

Author's response to reviews:

Thank you for the review of our manuscript (Rationale and Design of the CAPAMIS Study: Effectiveness of Pneumococcal Vaccination against Community-Acquired Pneumonia, Acute Myocardial Infarction and Stroke. MS: 1521217073244882). We have made all formatting changes suggested us by the editorial production team.

Bellow we listed these changes:

1. We modified the tables 1 and 2 as you suggested.
2. We added an Additional files section after the reference list.
3. Minor revisions: We change the sentence case in the title, remove some words from title page and made the changes in the abstract, references and figures that you suggested.

With best wishes,

Dra Olga Ochoa-Gondar
Corresponding author